Author's response to reviews

Title: Differential modulation of nicotine-induced gemcitabine resistance by GABA receptor agonists in pancreatic cancer xenografts and in vitro

Authors:

Jheelam Banerjee (jbanerje@utk.edu)
Hussein A.N. Al-Wadei (halwadei@vols.utk.edu)
Mohammed H. Al-Wadei (mohammedh2010@gmail.com)
Koami Dagnon (dagnon.clement@gmail.com)
Hildegard M. Schuller (hmsch@utk.edu)

Version: 4
Date: 19 August 2014

Author's response to reviews:

August 19, 2014

MS: 7320569111270591

Dear Editor:

In accordance with your request, we have now included the explanations about IACUC regulations that influenced experimental design on page 8 lines 1 through 10.

We hope that our manuscript will now be accepted for publication.

Sincerely,

Hildegard M. Schuller, DVM, Ph.D.
Distinguished Professor of Comparative Pathology